Author's response to reviews

Title: Termination of pregnancy in thalassemia major, hemophilia and Down's syndrome; what Iranian Physicians think?

Authors:

Mehran Karimi (karimim@sums.ac.ir)
Mohammad Mehdi Bonyadi (Bonyadim@sums.ac.ir)
Mohhamad reza Galehdari (galehdari@sums.ac.ir)
Soheila Zareifar (zareifars@sums.ac.ir)

Version: 2 Date: 6 April 2008

Author's response to reviews:

To: BMC Journal of Medical Ethics,
Sub: Submission of Manuscript for publication

Dear Editor-in-chief of BMC Journal of Medical Ethics,

Many thank for your kind cooperation. we received the edition of the article entitled" Termination of Pregnancy due to Thalassemia Major, Hemophilia, and Down's Syndrome: The Views of Iranian Physicians" from International Science Editing Yesterday.

So, we did your recommendations as Bellow:
1. We added more details to the method section and prepared a questioner.
2. We send you the edition which we received from International Science Editing. All Parts of the abstract and manuscript were revised as you Wish.

I would be thankful if it is processed at your earliest convenience. I thank you in advance for your time and efforts. Looking forward to hearing from you soon.

Thanking you,
Yours’ sincerely,
Mehran Karimi MD,
Hemostasis and Thrombosis,
Hematology Research Center,
Nemazee Hospital,
Shiraz University of Medical Sciences,
Tel & Fax: 711 6470207
E-mail: karimim@sums.ac.ir